Author's response to reviews

Title: A positioning pillow to improve lumbar puncture in paediatric haematology-oncology patients: A randomized controlled trial.

Authors:

Perrine Marec-Bérard (marec@lyon.fnclcc.fr)
Alvine Bissery (alvine.bissery@chu-lyon.fr)
Kamila Kebaïli (kamila.kebaïli@chu-lyon.fr)
Matthias Schell (schell@lyon.fnclcc.fr)
Françoise Aubert (francoise.aubert@chu-lyon.fr)
Ségolène Gaillard (se golene.gaillard@chu-lyon.fr)
Muriel Rabilloud (muriel.rabilloud@chu-lyon.fr)
Behrouz Kassaï (bk@upcl.univ-lyon1.fr)
Catherine Cornu (catherine.cornu@chu-lyon.fr)

Version: 8 Date: 13 January 2009

Author's response to reviews:

Corresponding author was changed
LPs was corrected in LP in abstract
page 5 A space was added before Four
page 6 a ) was removed, prvious LP was corrected to previous LPs
page 9 line 6, are results was corrected to our results
page 10 LPs was corrected to LP
page 12 of was removed after conceived
page 13 spaces were removed before ;
Figure 1 legend a ? was removed
page 19 Outcome was corrected in outcomes in tables